Status in Laser and Mirror group

E. Hirose, N. Mio (absent today), Nov 8, 2011

Laser

- Not much progress since last f2f meeting
- Ordered laser modules (Mitsubishi), and fiber amplifiers (Nufern)
 - -> Mio meeting engineers from Nufern this Friday

Mirror

- MC polish is done -> looking for coaters
- BS is going to be polished/coated at CSIRO (has been discussed)
- Specifications/drawings are in progress (1st version will be done this week)
 - -> Thanks to Hiro Yamamoto-san, Somiya-san, Aso-san, Iwasaki-san
 - -> Start discussion with some companies
- Visited Canon to discuss polishing/coating last week (Ohashi, Mio, Hirose)
- LIGO mirrors (for RM) are at Narita airport -> arriving at ICRR this week
 - -> The rest are still in LHO and LLO
- Preparing some documents to order C-axis sapphire